

Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL088039

Key Points:

- Krill hotspot formation and persistence are strongly heterogeneous in space and time
- Krill hotspots are shaped by local upwelling variability and regional circulation
 Krill hotspots coincide with
- observed distributions of marine mammals and seabirds

Supporting Information:

Supporting Information S1

Correspondence to: J. Fiechter, fiechter@ucsc.edu

Citation:

Fiechter, J., Santora, J. A., Chavez, F., Northcott, D., & Messié, M. (2020). Krill hotspot formation and phenology in the California Current Ecosystem. *Geophysical Research Letters*, 47, e2020GL088039. https://doi.org/ 10.1029/2020GL088039

Received 21 MAR 2020 Accepted 4 JUN 2020 Accepted article online 9 JUN 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Krill Hotspot Formation and Phenology in the California Current Ecosystem

Jerome Fiechter¹, Jarrod A. Santora^{2,3}, Francisco Chavez⁴, Devon Northcott^{4,5}, and Monique Messié⁴

¹Ocean Sciences Department, University of California, Santa Cruz, CA, USA, ²Fisheries Ecology Division, Southwest Fisheries Science Center, National Marine Fisheries Service, National Oceanic and Atmospheric Administration, Santa Cruz, CA, USA, ³Department of Applied Mathematics, University of California, Santa Cruz, CA, USA, ⁴Monterey Bay Aquarium Research Institute, Moss Landing, CA, USA, ⁵Now at Scripps Institution of Oceanography, University of California, San Diego, La Jolla, CA, USA

Abstract In the California Current Ecosystem, krill represent a key link between primary production and higher trophic level species owing to their central position in the food web and tendency to form dense aggregations. However, the strongly advective circulation associated with coastal upwelling may decouple the timing, occurrence, and persistence of krill hotspots from phytoplankton biomass and nutrient sources. Results from a coupled physical-biological model provide insights into fundamental mechanisms controlling the phenology of krill hotspots in the California Current Ecosystem, and their sensitivity to alongshore changes in coastal upwelling intensity. The simulation indicates that dynamics controlling krill hotspot formation, intensity, and persistence on seasonal and interannual timescales are strongly heterogeneous and related to alongshore variations in upwelling-favorable winds, primary production, and ocean currents. Furthermore, regions promoting persistent krill hotspot formation coincide with increased observed abundance of top predators, indicating that the model resolves important ecosystem complexity and function.

Plain Language Summary The California Current region off the west coast of North America is one of the four main upwelling systems of the World Ocean, and it supports a productive marine ecosystem including economically important fisheries. A numerical model representing the effects of ocean circulation on plankton dynamics is applied to explain how locally enhanced nutrient supply and transport by nearshore currents creates regions favoring the aggregation of krill, a key zooplankton species responsible for transferring energy between the base of the food web and marine predators. The two main conclusions from the study are that (1) the location, timing, and intensity of krill aggregations vary greatly both seasonally and from year to year and (2) regions of enhanced krill abundances in the model coincide with an increased observed presence of whales and seabirds, which suggest that these aggregations may support broader ecosystem hotspots.

1. Introduction

In marine ecosystems, linkages between planktonic organisms, forage fish, and predators must be considered in a context where ocean currents continuously reshape and rescale the underlying physical and biological environments. While, in theory, bottom-up processes influencing local nutrient enrichment and retention lead to sustained aggregations of mid and high trophic level species (Bakun, 1996; Santora et al., 2017), the formation and persistence of ecosystem hotspots in regions dominated by a strongly advective ocean circulation may become decoupled in space and time from nutrient sources and primary production (Messié & Chavez, 2017), making it difficult to understand trophic transfer of energy and predict aggregation locations.

In the California Current Ecosystem (CCE), one of the four major eastern boundary current upwelling regions, zooplankton species represent a key link between primary production and higher trophic level species. In particular, krill (euphausiids) hotspots, defined here as reoccurring areas of patchy high concentrations that range in size from tens to hundreds of kilometers and persist over weeks to months, often coincide with increased abundance of higher trophic level organisms, including commercially harvested species

(Santora et al., 2011; Santora, Dorman, et al., 2017; Santora, Sydeman, et al., 2017). Furthermore, growth and reproduction of krill in the CCE are linked to seasonal variability of upwelling (i.e., the primary mechanism supplying nutrients that support primary production) and advective properties of the ocean circulation (Brinton, 1962b; Brinton & Townsend, 2003; Croll et al., 2005; Messié & Chavez, 2017). Subsequent aggregation and retention of these organisms are influenced by local physical processes and topographic features (Bakun, 1996; Santora et al., 2018; Woodson & Litvin, 2014), thereby mediating the potential formation of ecosystem hotspots by controlling their distribution and availability to higher trophic level species. During the upwelling season, the dominant circulation features controlling advective properties in the CCE are the strong alongshore (equatorward) coastal jet associated with nearshore upwelling, mesoscale processes in the transition zone, and the slower meandering flow of the California Current proper offshore (Checkley & Barth, 2009).

While satellites have greatly expanded the ability to concurrently observe atmospheric conditions, surface physics, and primary production over a wide range of spatial and temporal scales (Edwards et al., 2010), they provide limited information on the occurrence, structure, and underlying biophysical processes associated with ecosystem hotspots involving midwater forage species. This is especially true of pelagic ecosystems where foundational species, such as zooplankton and schooling forage fish, form dense aggregations and have population dynamics intensely subject to climate variability (Chavez et al., 2003; Lindegren et al., 2013; Rykaczewski & Checkley, 2008). Coupled biophysical models, combined with observational studies, offer a promising way forward to elucidate mechanisms associated with ecosystem hotspot formation, phenology, and sensitivity to climate change (Cury et al., 2008). In the CCE, recent modeling approaches, parameterized and evaluated with existing observations, have been used to assess environmental drivers of variability in various planktonic organisms (Dorman, Sydeman, Garcia-Reyes, Zeno, & Santora, 2015; Fiechter et al., 2018; Messié & Chavez, 2017; Santora, Dorman, & Sydeman, 2017).

This study expands on previous modeling work by revealing how scales of physical advection combine with biological responses to sustain seasonal formation and persistence of krill hotspots, and their connections to alongshore variation in coastal upwelling intensity. Importantly, the model solution is compared with fisheries ecosystem surveys that monitor krill fluctuations and distribution patterns of top predators, leading to broader considerations on the ecosystem implications of resolving seasonal krill hotspots in the CCE and other krill centric systems.

2. Methods

The numerical model is an implementation of the Regional Ocean Modelling System (ROMS; Haidvogel et al., 2008; Shchepetkin & McWilliams, 2005) coupled to NEMUCSC, a customized version of the North Pacific Ecosystem Model for Understanding Regional Oceanography (NEMURO; Fiechter et al., 2018; Kishi et al., 2007). NEMUCSC includes three limiting macronutrients (nitrate, ammonium, and silicic acid), two phytoplankton (nanophytoplankton and diatoms) and three zooplankton (micro, meso, and predatory) functional groups, and three detritus pools (dissolved and particulate organic nitrogen and particulate silica). The predatory zooplankton group is parameterized to represent *Euphausia pacifica*, the numerically dominant euphausiid in the CCE (Brinton, 1962a, 1962b; Lavaniegos & Ohman, 2007). While NEMURO originally used "predatory zooplankton" as a broad group including gelatinous plankton and euphausiids with uniform grazing rates (0.2 day⁻¹ at 0°C) on diatom, microzooplankton, and meso-zooplankton, NEMUCSC introduces traits specific to *Euphausia pacifica* by favoring predation on diatoms (0.3 day⁻¹ at 0°C) and copepods (0.1 day⁻¹ at 0°C; Dorman, Sydeman, Bograd, & Powell, 2015; Ohman, 1984; Table S1 in the supporting information). The reduced direct grazing mortality by predatory zooplankton on microzooplankton and mesozooplankton is accounted for by a fractional increase of their density-dependent natural mortality.

To improve the representation of regional circulation patterns, the ROMS model downscales a $1/10^{\circ}$ (~10 km) data-assimilative physical reanalysis for the broader CCE to a $1/30^{\circ}$ (~3 km) domain for the central CCE (Figure S1 in the supporting information; Fiechter et al., 2018). The $1/30^{\circ}$ model is run for 1990–2010, and the analysis focuses on nearshore (0–100 km offshore) variability during peak upwelling season in the central CCE (May–August). Simulated krill concentrations are evaluated with in situ estimates of krill abundance from the NOAA Rockfish Recruitment and Ecosystem Assessment Survey (RREAS; Figure S1):

relative abundance is measured as catch-per-unit-effort from midwater trawls during May–June 1990–2010, and krill aggregations are determined from concurrent acoustic surveys (starting in 2000), indexed as the nautical area scattering coefficient (m^2/nmi^2 ; Santora et al., 2011, 2018). Additional evaluation of physical and biogeochemical fields from the 1/10° and 1/30° model solutions is available in Schroeder et al. (2014), Neveu et al. (2016), and Fiechter et al. (2018).

Hotspots in the model are identified as local alongshore peaks in krill abundance based on zonally averaged (0–100 km offshore) values, limited yearly to periods when monthly mean concentrations exceed the annual mean plus one standard deviation (Santora et al., 2018; Dorman, Sydeman, Garcia-Reyes, et al., 2015). To provide greater ecological context, the following hotspot properties are subsequently defined annually: "intensity" (maximum krill concentrations), "duration" (number of months during which krill concentrations). These properties are calculated over 0.5° alongshore bands centered near the peaks identified using the method described above.

Time series of annual hotspot property (intensity, duration, or peak timing) averaged over each 0.5° alongshore bands are combined into an empirical orthogonal function (EOF) analysis (e.g., Thomson & Emery, 2014) to detect coherent modes of variability across hotspots (EOF spatial modes). The standardized temporal amplitudes of the EOF modes are subsequently used to identify environmental conditions associated with latitudinal hotspot response by contrasting years of positive amplitude (>0.5 standard deviation) and negative amplitude (< -0.5 standard deviation; Figure S2). The environmental variables considered are annual mean alongshore wind stress, vertical velocity at 40 m, depth of 26.0 isopycnal, and nitrate concentration at 60 m during May–August. For example, anomalous vertical velocities for a given EOF mode are calculated by subtracting vertical velocities averaged over negative amplitude years (< -0.5 SD) from those averaged over positive amplitude years (>0.5 SD).

3. Results

3.1. Patterns and Variability of Krill Aggregations

During May–June, when in situ krill observations are most consistently available, simulated krill concentrations resolve three distinct accumulation regions: two stronger ones near 35 and 37°N and a weaker one near 40°N. These regions are largely consistent with krill aggregations identified from acoustic surveys (Santora et al., 2011, 2018), although the model underestimates relative intensity at 35°N and overestimates it at 37°N (Figure 1, upper panels). Nonetheless, simulated krill concentrations explain approximately half of the observed spatial variance, lending confidence that the simulation adequately represents alongshore regions favoring persistent krill aggregations. Importantly, variable survey effort within a region and year may impact the assessment of observed krill aggregations estimated from acoustic sampling.

Low-frequency (3-year running mean) variability of simulated krill concentrations in the region 36-38°N (approximately Monterey Bay to Gulf of the Farallones) is comparable to observed relative abundance from midwater trawls and exhibits a periodicity of ~6 years and a long-term increasing trend (Figure 1, lower panels). However, the magnitude of the annual trend is underestimated by the model with an increase of 1.4% of the long-term mean compared to 5.0% for the observations. Interannual variations are also weaker in the model, owing to the stabilizing effect of the quadratic natural mortality term in NEMUCSC, which prevents exponential fluctuations in phytoplankton and zooplankton biomasses (viz. linear vs. logarithmic y axes in Figure 1, lower panels). Year-to-year differences are also attributable to inconsistencies between the simulation and in situ sampling; such as those arising from krill patchiness and relative contributions of other krill species (e.g., Thysanoessa spinifera) to total abundance. Since the midwater trawl catch-perunit-effort only provides a measure of relative abundance, simulated concentrations are further evaluated with springtime krill abundance reported by Lavaniegos and Ohman (2007) for central California (Figure S1). Depth-integrated (surface to 200-m depth) krill concentrations in the model average to 877 mgC/m^2 during 1990–2010, compared to a mean observed euphausiid abundance of 294 mgC/m² during 1951-2005 (NEMUCSC nitrogen units are converted to carbon using a C:N of 4 (Lindley et al., 1999)). While the model overestimates the observed mean by a factor of 3, the discrepancy may be partly explained by the increasing trend in krill abundance during 1990-2010, combined with the fact that observations are biased toward 1951-1990 with only six annual springtime values available for 1990-2005.



Geophysical Research Letters



Figure 1. Spatial and temporal krill variability during May–June. Top: Simulated (left; $mmolN/m^3$) and observed (center; nautical area scattering coefficient *Z*-score) mean krill abundances during 2000–2010 at observation locations, and linear regression between simulated and observed values (right); contour lines indicate 100-m and 1,000-m isobaths. Bottom: Annual (left) and 3-year running mean (right) simulated (red; $mmolN/m^3$) and observed (blue; catch-per-unit-effort) krill abundances averaged over 36–38°N and 0–100 km offshore during 1990–2010 (running mean is applied to smooth out annual differences while retaining lower frequency variability); the dashed lines in left panel indicate long-term trends.

The limited spatiotemporal coverage of shipboard surveys off California restricts the ability to determine transient properties of enhanced krill concentration regions. Hence, the simulation is used to explore potential mechanisms controlling the phenology of krill aggregations in the central CCE in response to coastal upwelling variability. The most striking feature in the model solution is the alongshore variation in location and intensity of peak concentrations from May to August under the combined effects of alongshore advection by coastal currents and the seasonal northward progression of upwelling (Fiechter et al., 2018; Messié & Chavez, 2015; Figure 2). In the central and northern parts of the domain, elevated krill concentrations during spring are found near 37 and 40°N, which is approximately 100 km equatorward of where maximum phytoplankton biomasses occur (near 38 and 41°N, respectively), whereas elevated krill concentrations during summer are essentially collocated with regions of peak phytoplankton biomass (Figure S3).

The latitudinal offset between regions of simulated peak phytoplankton and krill biomasses corresponds to the distance local krill concentrations would typically be advected alongshore by near surface currents over a krill doubling period (~1 week in the model), which can reach 50–100 km during spring and decrease to less than 50 km during summer. This finding is dynamically consistent with the seasonal development of the alongshore coastal jet, which forms nearshore during spring and migrates offshore as upwelling strengthens (Strub & James, 2000), resulting in stronger alongshore advection near the coast in May–June relative to July–August. In contrast, peak phytoplankton and krill concentrations near 35°N remain collocated throughout the upwelling months, as alongshore transport in that region is typically weak with a poleward tendency (i.e., advection distances less than ~30 km on average; Figure S3). These latitudinal and seasonal changes in the spatial coupling between krill and their nutrient and prey sources further confirm the importance of surface advection as a key process shaping alongshore distribution of phytoplankton biomass (Fiechter et al., 2018) and krill aggregations (Dorman, Sydeman, Garcia-Reyes, et al., 2015; Messié & Chavez, 2017) in the CCE.





Figure 2. Phenology of krill hotspots. Left and center: Climatological mean simulated surface krill concentrations $(mmolN/m^3)$ during May, June, July, and August (the dashed contour line indicates 100 km offshore). Right: Simulated mean nearshore surface krill hotspot concentrations (red; mmolN/m³) and peak timing (blue; month) as a function of latitude; the shaded 0.5° bands denote alongshore extent of seasonal peaks identified as local hotspots in simulated surface krill concentrations averaged 0–100 km offshore.

3.2. Intensity, Timing, and Duration of Krill Hotspots

Over the course of the upwelling season, the simulation reveals the occurrence of five distinct hotspots: the three identified for May–June (which in fact represent four hotspots since the large concentration area near 37°N consists of two separate aggregation regions), plus one near 41°N (HS1–HS5 in Figure 2 and Table S2). The southernmost hotspot near 35°N (HS1) peaks the earliest (mid-May) and has the lowest intensity, but also the largest interannual variability (standard deviation) and annual trend. During 1990–2010, the trend at HS1 accounts for a cumulative increase in krill concentrations of approximately half the peak intensity and twice the standard deviation. While HS1 intensity has steadily increased, its duration has shortened over the same period by roughly half a month, which is comparable to the timescale of interannual variations. However, the decrease in duration is outpaced by the increase in peak intensity, leading to an overall increasing trend, albeit slower, in annual mean krill concentrations at HS1.

The large area of high krill concentrations in the central region (36.5–38°N) represents two distinct hotspots: one south of 37°N (HS2) and one north of 37°N (HS3). Although the two hotspots have similar means, standard deviations, and trends in their intensity, the subregion south of 37°N experiences peak krill concentrations 3 weeks earlier than the subregion north of 37°N (Table S2). Interannual shifts in their duration and peak timing are also largely uncorrelated. The increasing trend in intensity at HS2 and HS3 is smaller relative to the mean than at HS1, yet the cumulative effect of the trend over the 21-year period still exceeds interannual variability.

Two main aggregation regions occur in the northern part of the domain, the first peaking at 39.5–40°N during the first half of June (HS4) and the second reaching maximum intensity approximately a month later near 41°N (HS5). The earlier peak time of HS4 suggests that this hotspot may be sustained by northward advection of nearshore nutrients from the region of enhanced upwelling near 39°N (Fiechter et al., 2018). While both locations have similar mean intensities, interannual fluctuations in their properties are not strongly correlated and year-to-year intensity variation (standard deviation) near 41°N (HS5) is roughly



Geophysical Research Letters



Figure 3. Environmental patterns associated with krill hotspot variability. Alongshore empirical orthogonal function (EOF) spatial mode (far left) and corresponding anomalies in (from left to right) meridional wind stress (N/m^2) , vertical velocity at 40 m (m/day), depth of 26.0 isopycnal (m), and nitrate at 60 m (mmolN/m³) for (from top to bottom) first and second EOF modes for intensity (67 and 15% of explained variance), first EOF mode for peak timing (21% of explained variance), and first EOF mode for duration (32% of explained variance). The dashed contour line indicates 100 km offshore.

twice that at 39.5–40°N (HS4). However, HS4 displays the largest year-to-year variations in the timing of peak intensity among all five hotspots (standard deviation of ~1 month). Although annual mean duration is rather uniform across all hotspots and ranges between 2 and 2.5 months, the northernmost hotspot (HS5) tends to be most persistent (Table S2).

3.3. Environmental Variability Associated With Krill Hotspots

The disparate range of responses (i.e., means, trends, and interannual variability) across hotspots suggests that dynamics controlling hotspot formation are strongly heterogeneous along the coast. Hence, biophysical processes determining the duration, intensity, and timing of krill aggregations must be understood locally in the context of regional patterns. To that effect, an EOF analysis is used to isolate dominant patterns of along-shore krill hotspot variability and to identify specific environmental conditions associated with these modes.

The leading mode of variability for hotspot intensity (67% of explained variance) is a synchronous change (i.e., increase or decrease) in krill concentrations at all five alongshore locations (Figure 3, top row), with the magnitude of the response intensified in the southern part of the domain. Not unexpectedly, periods



Figure 4. Ecosystem hotspots in the California Current Ecosystem. Top left: Simulated krill aggregations and variability in their timing and intensity during 1990–2010; circle size and color indicate relative mean (M), variance (V), and trend (T) in intensity across locations. Bottom: Observed distributions of humpback whale (HUWH) (left) and sooty shearwater (SOSH) (center) during May–June 1996–2018. Right: Zonally averaged (0–60 km offshore) krill concentrations (red; mmolN/m³) and predator abundances (blue = HUWH; green = SOSH) normalized by their meridional (34–40°N) mean.

of increased hotspot intensity coincide with a region-wide strengthening of upwelling favorable alongshore winds, shallower isopycnals, and enhanced delivery of nutrients near the coast. Based on the temporal amplitude of the first EOF mode, krill hotspots in the CCE were notably more intense in 1991 and 2008 and weaker in 1992, 1997–1998, 2005–2006, and 2009 (Figure S3).

The second mode of variability for hotspot intensity (15% of explained variance) is an asynchronous alongshore response, where the regions north and south of Point Arena (~39°N) vary out of phase with maximum amplitudes at the southern and northern most latitudes (i.e., near 35 and 41°N; Figure 3, second row). This pattern is predominantly associated with changes in surface wind stress in the northern part of the domain, resulting in alongshore differences in isopycnal depths and nutrient concentrations. Reduced upwelling in the north (i.e., weaker alongshore winds) leads to relatively deeper isopycnals near the coast north of ~39°N, with a corresponding alongshore nutrient gradient at 60-m depth. The combination of weaker vertical velocities and lower nutrient concentrations below the mixed layer acts to limit production at the northern hotspot locations compared to the central and southern ones. Based on the temporal amplitude of the second EOF mode, krill hotspots in the southern CCE were notably more intense in 1993, 1997, and 2010 and weaker in 1998 and 2000–02 relative to krill hotspots in the northern CCE (Figure S4).

The dominant EOF mode for the timing of peak krill concentrations (21% of explained variance) exhibits low-frequency variability correlated to the second EOF mode for hotspot intensity and, consequently, both modes are associated with similar patterns of environmental variability. However, while alongshore variations in intensity are out of phase, those for peak timing are synchronous throughout the region

(Figure 3, third row). These results suggest that interannual variations in peak timing are related to those in hotspot intensity, and years for which peak krill concentrations occur later in the upwelling season coincide with an increase in hotspot intensity in the southern part of the domain and decrease in the northern part.

While the leading modes of variability for hotspot intensity and peak timing are both synchronous alongshore, that for duration (32% of explained variance) indicates an out-of-phase response between the southernmost hotspot near 35°N and the northern ones near 39.5–40°N and ~41°N (Figure 3, bottom row). Since anomalies in isopycnal depths and subsurface nutrient concentrations are synchronous coast-wide, this pattern is predominantly associated with an alongshore gradient in upwelling favorable winds, enhancing coastal upwelling intensity in the northern part of the domain and reducing it between 34 and 36°N.

4. Discussion

Observing and predicting the formation and persistence of krill and higher trophic level species hotspots is difficult, and the results presented here offer insight into fundamental processes shaping lower trophic level ecosystem response in the CCE across space and time. Not only does the model recreate the location of known krill hotspots and their relationship to upwelling centers but it also explains their spatial variability and phenological properties in relation to alongshore variation in primary productivity and advection by near-surface ocean currents (Figure 4, top left panel). The simulation further indicates that the magnitude of interannual variability and long-term trend decreases with latitude, with the southernmost hotspot (HS1) experiencing the most extreme values. This finding is consistent with analyzed variability from satellite altimetry demonstrating that the southern CCE responds predominantly on interannual timescales and the northern CCE on intraseasonal timescales (Chavez & Messié, 2009).

The low-frequency variability and long-term trend in simulated krill concentrations are generally supported by ecosystem assessment surveys, which found that krill abundances were significantly lower throughout the 1990s (characterized by a positive PDO phase and strong El Niño events in 1992 and 1997–1998) and overall higher during the cooler decade of the 2000s (Ralston et al., 2015; Santora et al., 2014; Sydeman et al., 2013, 2015). Furthermore, the region-wide decline in hotspot intensity predicted by the model during 2005–2007 corresponds to a previously identified period of delayed upwelling and intrusions of warm oceanic water, with known impacts to krill predators, such as reproductive failure of seabirds, distribution shifts of baleen whales and poor ocean salmon survival, and later recruitment (Fiechter et al., 2015; Fleming et al., 2006; Lindley et al., 2009; Sydeman et al., 2006).

Independent observations of the abundance and distribution of sooty shearwaters (*Ardenna grisea*) and humpback whales (*Megaptera novaeangliae*) collected during synoptic ecosystem surveys in May–June 1996–2018 closely resemble simulated patterns of krill hotspots within the central CCE (notably near 35°N and 36–38°N), which not only lends further confidence in the model's ability to reproduce krill aggregation locations but also suggests that these hotspots are indicative of broader ecosystem phenomena (Figure 4). The structural relationship between simulated krill aggregations and observed seabird and whale distributions implies that the model may be valuable for informing regional productivity assessments and ecosystem-based management (PFMC, 2008; Lester et al., 2010; Maxwell et al., 2015), including linkages to populations of krill predators that are either resident breeders or migrate throughout the CCE in search of krill hotspots (Adams et al., 2012; Bailey et al., 2009; Calambokidis et al., 2015; Irvine et al., 2014; Redfern et al., 2013).

Finally, the modeling framework presented here can be extended to other krill-centric marine ecosystems to facilitate a cross-regional evaluation of processes governing krill hotspot formation and persistence. For instance, since local alongshore topographic impacts on upwelling intensity and productivity modulated by regional circulation patterns and basin-scale influences are ubiquitous features of eastern boundary current upwelling regions, similar considerations on fundamental processes controlling krill aggregations and ecosystem hotspots should equally apply to the Benguela, Canary, and Humboldt Current Systems.

References

Adams, J., MacLeod, C., Suryan, R. M., Hyrenbach, K. D., & Harvey, J. T. (2012). Summer-time use of west coast US National Marine Sanctuaries by migrating sooty shearwaters (Puffinus griseus). *Biological Conservation*, 156, 105–116. https://doi.org/10.1016/j. biocon.2011.12.032

Acknowledgments

This study was primarily funded by a grant from the National Aeronautics and Space Administration (80NSSC17K0574), with additional support from two awards from the National Science Foundation (OCE1566623 and OCE1635315) and from the David and Lucile Packard Foundation. The authors thank the NOAA NMFS Southwest Fisheries Science Center and dedicated researchers maintaining RREAS. The Farallon Institute is acknowledged for providing seabirds and marine mammal observations collected during RREAS. The authors also thank two anonymous reviewers for their constructive comments. The model output is deposited on Dryad at https:// doi.org/10.7291/D1KD4J. RREAS data are available at https://coastwatch.pfeg. noaa.gov/erddap/tabledap/FED_ Rockfish_Catch.html.

Bailey, H., Mate, B. R., Palacios, D. M., Irvine, L., Bograd, S. J., & Costa, D. P. (2009). Behavioural estimation of blue whale movements in the Northeast Pacific from state-space model analysis of satellite tracks. *Endangered Species Research*, 10, 93–106. https://doi.org/ 10.3354/esr00239

Bakun, A. (1996). Patterns in the ocean: Ocean processes and marine population dynamics. San Diego: California Sea Grant College System. Brinton, E. (1962a). The distribution of Pacific euphausiids. Bulletin of the Scripps Institution of Oceanography of the University of California, La Jolla, 8, 51–269.

- Brinton, E. (1962b). Variable factors affecting the apparent range and estimated concentration of euphausiids in the North Pacific *Pacific Science*, *16*, 374–408.
- Brinton, E., & Townsend, A. (2003). Decadal variability in abundances of the dominant euphausiid species in southern sectors of the California Current. Deep Sea Research Part II: Topical Studies in Oceanography, 50(14–16), 2449–2472. https://doi.org/10.1016/S0967-0645(03)00126-7
- Calambokidis, J., Steiger, G. H., Curtice, C., Harrison, J., Ferguson, M. C., Becker, E., et al. (2015). Biologically important areas for selected cetaceans within U.S. waters-West Coast Region. *Aquatic Mammals*, 41(1), 39–53. https://doi.org/10.1578/AM.41.1.2015.39
- Chavez, F. P., & Messié, M. (2009). A comparison of eastern boundary upwelling ecosystems. *Progress in Oceanography*, 83(1–4), 80–96. https://doi.org/10.1016/j.pocean.2009.07.032
- Chavez, F. P., Ryan, J. P., Lluch-Cota, S. E., & Niquen, M. (2003). From anchovies to sardines and back: Multidecadal change in the Pacific Ocean. Science, 299(5604), 217–221. https://doi.org/10.1126/science.1075880
- Checkley, D. M., & Barth, J. A. (2009). Patterns and processes in the California Current System. Progress in Oceanography, 83(1-4), 49–64. https://doi.org/10.1016/j.pocean.2009.07.028
- Croll, D. A., Marinovic, B., Benson, S., Chavez, F. P., Black, N., Ternullo, R., & Tershy, B. R. (2005). From wind to whales: Trophic links in a coastal upwelling system. *Marine Ecology Progress Series*, 289, 117–130. https://doi.org/10.3354/meps289117
- Cury, P., Shin, Y.-J., Planque, B., Durant, J. M., Fromentin, J.-M., Kramer-Schadt, S., et al. (2008). Ecosystem oceanography for global change in fisheries. *Trends in Ecology & Evolution*, 23(6), 338–346. https://doi.org/10.1016/j.tree.2008.02.005
- Dorman, J. G., Sydeman, W. J., Bograd, S. J., & Powell, T. M. (2015). An individual-based model of the krill Euphausia pacifica in the California Current. *Progress in Oceanography*, *138*, 504–520. https://doi.org/10.1016/j.pocean.2015.02.006
- Dorman, J. G., Sydeman, W. J., Garcia-Reyes, M., Zeno, R. L., & Santora, J. A. (2015). Modeling krill aggregations in the central-northern California Current. Marine Ecology Progress Series, 528, 87–99. https://doi.org/10.3354/meps11253
- Edwards, M., Beaugrand, G., Hays, G. C., Koslow, J. A., & Richardson, A. J. (2010). Multi-decadal oceanic ecological datasets and their application in marine policy and management. *Trends in Ecology & Evolution*, 25(10), 602–610. https://doi.org/10.1016/j. tree.2010.07.007
- Fiechter, J., Edwards, C. A., & Moore, A. M. (2018). Wind, circulation, and topographic effects on alongshore phytoplankton variability in the California Current. *Geophysical Research Letters*, 45, 3238–3245. https://doi.org/10.1002/2017GL076839
- Fiechter, J., Huff, D. D., Martin, B. T., Jackson, D. W., Edwards, C. A., Rose, K. A., et al. (2015). Environmental conditions impacting juvenile Chinook salmon growth off central California: An ecosystem model analysis. *Geophysical Research Letters*, 42, 2910–2917. https://doi.org/10.1002/2015GL063046
- Fleming, A. H., Clark, C. T., Calambokidis, J., & Barlow, J. (2016). Humpback whale diets respond to variance in ocean climate and ecosystem conditions in the California Current. Global Change Biology, 22(3), 1214–1224. https://doi.org/10.1111/gcb.13171
- Haidvogel, D. B., Arango, H., Budgell, W. P., Cornuelle, B. D., Curchitser, E., di Lorenzo, E., et al. (2008). Ocean forecasting in terrain-following coordinates: Formulation and skill assessment of the Regional Ocean Modeling System. *Journal of Computational Physics*, 227(7), 3595–3624. https://doi.org/10.1016/j.jcp.2007.06.016
- Irvine, L. M., Mate, B. R., Winsor, M. H., Palacios, D. M., Bograd, S. J., Costa, D. P., & Bailey, H. (2014). Spatial and temporal occurrence of blue whales off the U.S. west coast, with implications for management. *PLoS ONE*, 9(7), e102959. pmid:25054829. https://doi.org/ 10.1371/journal.pone.0102959
- Kishi, M. J., Kashiwai, M., Ware, D. M., Megrey, B. A., Eslinger, D. L., Werner, F. E., et al. (2007). NEMURO—A lower trophic level model for the North Pacific marine ecosystem. *Ecological Modelling*, 202(1-2), 12–25. https://doi.org/10.1016/j.ecolmodel.2006.08.021
- Lavaniegos, B. E., & Ohman, M. D. (2007). Coherence of long-term variations of zooplankton in two sectors of the California Current System. Progress in Oceanography, 75(1), 42–69. https://doi.org/10.1016/j.pocean.2007.07.002
- Lester, S. E., McLeod, K. L., Tallis, H., Ruckelshaus, M., Halpern, B. S., Levin, P. S., et al. (2010). Science in support of ecosystem-based management for the US West Coast and beyond. *Biological Conservation*, 143(3), 576–587. https://doi.org/10.1016/j.biocon.2009.11.021
- Lindegren, M., Checkley, D. M., Rouyer, T., MacCall, A. D., & Stenseth, N. C. (2013). Climate, fishing, and fluctuations of sardine and anchovy in the California Current. *Proceedings National Academy Sciences USA*, 110(33), 13,672–13,677. https://doi.org/10.1073/ pnas.1305733110
- Lindley, J. A., Robbins, D. B., & Williams, R. (1999). Dry weight carbon and nitrogen content of some euphausiids from the North Atlantic Ocean and the Celtic Sea. Journal of Plankton Research, 21(11), 2053–2066. https://doi.org/10.1093/plankt/21.11.2053
- Lindley, S. T., Grimes, C. B., Mohr, M. S., Peterson, W., Stein, J., Anderson, J. T., et al. (2009). What caused the Sacramento River fall Chinook stock collapse? NOAA Tech. Memo. NOAA-TM-NMFS-SWFSC-447.
- Maxwell, S. M., Hazen, E. L., Lewison, R. L., Dunn, D. C., Bailey, H., Bograd, S. J., et al. (2015). Dynamic ocean management: Defining and conceptualizing real-time management of the ocean. *Marine Policy*, 58, 42–50. https://doi.org/10.1016/j.marpol.2015.03.014
- Messié, M., & Chavez, F. P. (2015). Seasonal regulation of primary production in eastern boundary upwelling systems. Progress in Oceanography, 134, 1–18. https://doi.org/10.1016/j.pocean.2014.10.011
- Messié, M., & Chavez, F. P. (2017). Nutrient supply, surface currents, and plankton dynamics predict zooplankton hotspots in coastal upwelling systems. *Geophysical Research Letters*, 44, 8979–8986. https://doi.org/10.1002/2017GL074322
- Neveu, E., Moore, A. M., Edwards, C. A., Fiechter, J., Drake, P., Jacox, M. G., & Nuss, E. (2016). An historical analysis of the California Current using ROMS 4D-Var: System configuration and diagnostics. *Ocean Modelling*, 99, 131–151. https://doi.org/10.1016/j.ocemod. 2015.11.012
- Ohman, M. D. (1984). Omnivory by Euphausia pacifica: The role of copepod prey. Marine Ecology Progress Series, 19, 25–131.
- Pacific Fisheries Management Council. (2008). Management if krill as an essential component of the California Current Ecosystem, http://www.pcouncil.org/wp-content/uploads/CPS_Am12_Krill_DraftEA.pdf
- Ralston, S., Field, J. C., & Sakuma, K. M. (2015). Long-term variation in a central California pelagic forage assemblage. *Journal of Marine Systems*, 146, 26–37. https://doi.org/10.1016/j.jmarsys.2014.06.013
- Redfern, J. V., McKenna, M. F., Moore, T. J., Calambokidis, J., Deangelis, M. L., Becker, E. A., et al. (2013). Assessing the risk of ship striking large whales in marine spatial planning. *Conservation Biology*, *27*(2), 292–302. https://doi.org/10.1111/cobi.12029

- Rykaczewski, R. R., & Checkley, D. M. (2008). Influence of ocean winds on the pelagic ecosystem in upwelling regions. Proceedings National Academy of Sciences USA, 105(6), 1965–1970. https://doi.org/10.1073/pnas.0711777105
- Santora, J. A., Dorman, J. G., & Sydeman, W. J. (2017). Modeling spatiotemporal dynamics of krill aggregations: Size, intensity, persistence, and coherence with seabirds. *Ecography*, 40(11), 1300–1314. https://doi.org/10.1111/ecog.02250
 - Santora, J. A., Schroeder, I. D., Field, J. C., Wells, B. K., & Sydeman, W. J. (2014). Spatio-temporal dynamics of ocean conditions and forage taxa reveal regional structuring of seabird-prey relationships. *Ecological Applications*, 24(7), 1730–1747. https://doi.org/10.1890/13-1605.1
- Santora, J. A., Sydeman, W. J., Schroeder, I. D., Field, J. C., Miller, R. R., & Wells, B. K. (2017). Persistence of trophic hotspots and relation to human impacts within an upwelling marine ecosystem. *Ecological Applications*, 27(2), 560–574. https://doi.org/10.1002/eap.1466
- Santora, J. A., Sydeman, W. J., Schroeder, I. D., Wells, B. K., & Field, J. C. (2011). Mesoscale structure and oceanographic determinants if krill hotspots in the California Current: Implications for trophic transfer and conservation. *Progress in Oceanography*, 91(4), 397–409. https://doi.org/10.1016/j.pocean.2011.04.002
- Santora, J. A., Zeno, R., Dorman, J. G., & Sydeman, W. J. (2018). Submarine canyons represent an essential habitat network for krill hotspots in a large marine ecosystem. *Scientific Reports*, 8(1), 7579. https://doi.org/10.1038/s41598-018-25742-9
- Schroeder, I. D., Santora, J. A., Moore, A. M., Edwards, C. A., Fiechter, J., Hazen, E. L., et al. (2014). Application of a data-assimilative regional ocean modeling system for assessing California Current System ocean conditions, krill, and juvenile rockfish interannual variability. *Geophysical Research Letters*, 41, 5942–5950. https://doi.org/10.1002/2014GL061045
- Shchepetkin, A. F., & McWilliams, J. C. (2005). The regional oceanic modeling system (ROMS): A split-explicit, free-surface, topographyfollowing-coordinate oceanic model. *Ocean Modelling*, 9(4), 347–404. https://doi.org/10.1016/j.ocemod.2004.08.002
- Strub, P. T., & James, C. (2000). Altimeter-derived variability of surface velocities in the California Current System: 2. Seasonal circulation and eddy statistics. Deep Sea Research Part II: Topical Studies in Oceanography, 47(5–6), 831–870. https://doi.org/10.1016/S0967-0645(99)00129-0
- Sydeman, W. J., Bradley, R. W., Warzybok, P., Abraham, C. L., Jahncke, J., Hyrenbach, K. D., et al. (2006). Planktivorous auklet Ptychoramphus aleuticus responses to ocean climate, 2005: Unusual atmospheric blocking? *Geophysical Research Letters*, 33, L22S09. https://doi.org/10.1029/2006GL026736
- Sydeman, W. J., Santora, J. A., Thompson, S. A., Marinovic, B., & Lorenzo, E. D. (2013). Increasing variance in North Pacific climate relates to unprecedented ecosystem variability off California. *Global Change Biology*, *19*(6), 1662–1675. https://doi.org/10.1111/gcb.12165
- Sydeman, W. J., Thompson, S. A., Santora, J. A., Koslow, J. A., Goericke, R., & Ohman, M. D. (2015). Climate-ecosystem change off southern California: Time-dependent seabird predator-prey numerical responses. *Deep-Sea Research Part II*, 112, 158–170. https://doi. org/10.1016/j.dsr2.2014.03.008
- Thomson, R. E., & Emery, W. J. (2014). Data analysis methods in physical oceanography. Newnes, Waltham, MA: Elsevier.
- Woodson, C. B., & Litvin, S. Y. (2014). Ocean fronts drive marine fishery production and biogeochemical cycling. Proceedings National Academy of Sciences USA. https://doi.org/10.1073/pnas.141743112